Author's response to reviews

Title: The efficacy of a brief intervention to reduce alcohol misuse in patients with HIV in South Africa: study protocol for a randomized controlled trial.

Authors:

Diana Huis in ’t Veld (dhuisintveld@itg.be)
Linda Skaal (Linda_Skaal@embnet.com)
Karl Peltzer (kpeltzer@hsrc.ac.za)
Robert Colebunders (bcoleb@itg.be)
John V Ndimande (ndimande@gpnet.net)
Supa Pengpid (Supa_Pengpid@embnet.com)

Version: 6 Date: 19 September 2012

Author’s response to reviews:

Dear reviewers,

The title of the manuscript has been adapted conform the journal style for study protocol articles to: The efficacy of a brief intervention to reduce alcohol misuse in patients with HIV in South Africa: study protocol for a randomized controlled trial.

The manuscript is checked according to the instructions for authors, the text below the headings: "Competing interests" and "Acknowledgements" has been adapted according to the instructions for authors.